

NOTA DI LAVORO

88.2009

SMEs in Argentina: Who are the Exporters

By Gianmarco I.P. Ottaviano, Bocconi University DEP-KITeS, FEEM and CEPR Christian Volpe Martincus, Inter-American Development Bank

GLOBAL CHALLENGES Series Editor: Gianmarco I.P. Ottaviano

SMEs in Argentina: Who are the Exporters

By Gianmarco I.P. Ottaviano, Bocconi University DEP-KITeS, FEEM and CFPR

Christian Volpe Martincus, Inter-American Development Bank

Summary

There exists a growing body of literature which looks at export decisions made by firms. Most studies focus on developed countries and do not explore whether different behavioral patterns prevail over the firm size distribution. This paper aims at filling this gap in the literature by analyzing the export behavior of a statistically representative sample of 192 Small and Medium-Size Enterprises (SMEs) in a developing country, Argentina, over the period 1996-1998. We find that the level of employment, sourcing from abroad, investment in product improvement and average productivity are associated with a higher probability of exporting. Training activities for employees are important to export outside of MERCOSUR.

Keywords: SME, Exports, Argentina

JEL Classification: F10, F14, D21, L60

We would like to thank our editor, Rui Baptista, and two anonymous referees for valuable comments and suggestions. We also wish to thank Vicente Donato for giving us access to the "Observatorio PyME" database collected by the University of Bologna at Buenos Aires. Finally, we owe gratitude to Jerónimo Carballo, Eduardo Mendoza Quintanilla, and Christian Haedo for their excellent research assistance and to Carlos Gutierrez Jr. for his valuable editing assistance. When the first draft was finalized, the first author belonged to the faculty of the University of Bologna. The views and interpretation in this document are strictly those of the authors and should not be attributed to the Inter-American Development Bank, its Executive Directors or its member countries. Other usual disclaimers also apply.

Address for correspondence:

Gianmarco I.P. Ottaviano Bocconi University Via Sarfatti 25 20136 Milan Italy

Phone: +39 5836 3385 Fax: +39 5836 3309

E-mail: gianmarco.ottaviano@unibocconi.it

SMEs in Argentina: Who are the Exporters?

Gianmarco I. P. Ottaviano*

Bocconi University, Dept. of Economics and
KITeS; FEEM and CEPR

Christian Volpe Martincus* Inter-American Development Bank

This version: June 2009

Abstract

There exists a growing body of literature which looks at export decisions made by firms. Most studies focus on developed countries and do not explore whether different behavioral patterns prevail over the firm size distribution. This paper aims at filling this gap in the literature by analyzing the export behavior of a statistically representative sample of 192 Small and Medium-Size Enterprises (SMEs) in a developing country, Argentina, over the period 1996-1998. We find that the level of employment, sourcing from abroad, investment in product improvement and average productivity are associated with a higher probability of exporting. Training activities for employees are important to export outside of MERCOSUR.

Keywords: SME, Exports, Argentina JEL-Codes F10, F14, D21, L60

[•] We would like to thank our editor, Rui Baptista, and two anonymous referees for valuable comments and suggestions. We also wish to thank Vicente Donato for giving us access to the "Observatorio PyME" database collected by the University of Bologna at Buenos Aires. Finally, we owe gratitude to Jerónimo Carballo, Eduardo Mendoza Quintanilla, and Christian Haedo for their excellent research assistance and to Carlos Gutierrez Jr. for his valuable editing assistance. When the first draft was finalized, the first author belonged to the faculty of the University of Bologna. The views and interpretation in this document are strictly those of the authors and should not be attributed to the Inter-American Development Bank, its Executive Directors or its member countries. Other usual disclaimers also apply.

^{*} Correspondence Address: Bocconi University, Dept. of Economics, Via Roentgen 1, 20136 Milan, Italy. E-mail: gianmarco.ottaviano@unibocconi.it. Tel: +39 5836 3385. Fax: +39 5836 3309.

^{*} Correspondence Address: Inter-American Development Bank, Stop W0610, 1300 New York Avenue, NW, Washington, DC 20577, United States of America. E-mail: christianv@iadb.org. Tel: +1 202 623 3199. Fax: +1 202 623 2995.

SMEs in Argentina: Who are the Exporters?

1 Introduction

There is a growing body of literature analyzing the export behavior of firms. Most studies look at developed countries and do not explicitly explore whether the corresponding findings remain robust across firm sizes. In particular, most firm-level dataset only include firms with size above a certain threshold. Thus, evidence on the lower segments of the size distribution in developing countries is rather scarce. This paper aims at filling this gap in the literature by examining the export decisions by Small and Medium Size Firms (SMEs) in Argentina. This is an interesting case study because, as a developing country, macroeconomic and trade policy environments are more volatile in Argentina than in a typical developed economy, thus different transition rates into and out of the export markets can be expected. Further, smaller firms are likely to be particularly affected by those conditions. It therefore seems to be worth exploring whether they have distinguishing export behavior patterns.

An already well established stylized fact in the empirical literature on international trade is that trade flows exhibit hysteresis. More specifically, firms' current foreign market participation is strongly influenced by prior experience, thus reflecting the existence of sunk costs in entering external markets. Hence, when assessing the factors that shape firms' entry and exit from these markets, it is necessary to properly account for these sunk costs. For this reason our analysis starts with the theoretical model of firms' export decision featuring sunk entry costs proposed by Roberts and Tybout (1997) and later used, among others, by Bernard and Jensen (1999, 2004). We then derive and empirically estimate binary choice models of firms' exporting behavior. In particular, we first estimate pooled probit regressions with both errors clustered by firm to deal with autocorrelation and bootstrapped standard errors. Successively, we turn to random-effects probit regressions to account for possible unobserved heterogeneity.

This kind of analysis allows us to uncover the factors associated with firms' participation to international markets. In addition, it is interesting to check whether firms' export behavior differs

¹ Bernard and Jensen (1999, 2004) examine the determinants of firms' export performance in the United States; Bernard and Wagner (2001) and Arnold and Hussinger (2005) in Germany; Castellani (2002) and Castellani et al. (2009) in Italy; Delgado et al. (2002), Barrios et al. (2003), and Blanes-Cristóbal et al. (2008) in Spain; Head and Ries (2003) in Japan; Eaton et al. (2004, 2008) in France; Girma et al. (2004), Greenaway et al. (2007), and Kneller et al. (2008) in the United Kingdom; and Lawless (2009) in Ireland.

² Alvarez (2004) examines determinants of export performance of Chilean manufacturing SMEs. He finds that greater effort in international business, process innovation, and use of export promotion programs are associated with higher probability of being a permanent exporter.

across specific destination countries. The reason is that it has been recently argued in the theoretical literature that the existence of differences in the toughness of competition across countries (as determined by market size and the extent of trade integration) translate into differences in the set of heterogeneous firms that are active across them (see Melitz and Ottaviano, 2008). Furthermore, sunk entry costs can differ among destination countries (see, e.g., Blanes-Cristóbal et al., 2008). For instance, the costs of gathering information on business opportunities abroad are likely to be smaller for nearby countries because commercial exchanges are more intense and media coverage is better (see, e.g., Grossman, 1998; Anderson, 2000; Portes et al., 2001; Loungani et al., 2002; Hwang, 2007; Guiso et al., 2008). The same would hold for countries that are partners in a preferential trade agreement. Lastly, different countries may demand goods of different quality (see, e.g., Hallak, 2006). For example, in the case of Argentine firms, exporting to specific destinations outside of Latin America, which are primarily distant developed countries, might be more likely if the quality of the workforce is improved via training and better or directly new products are introduced (see, e.g., Bernard and Jensen, 2004). Hence, we also estimate seemingly unrelated biprobit regressions to analyze export behavior across two destinations, the neighboring trade partners, which are also members of MERCOSUR (Brazil, Paraguay, and Uruguay) and represented the main destination for Argentine manufacturing exports over the sample period, and the rest of the world.4

In so doing, we exploit a new database collected by the University of Bologna's Representation in Argentina that provides information for a statistically representative sample of 192 Argentinean SMEs over the period 1996-1998. This database identifies both the location and the industry the firm belongs to (i.e., its main activity) and contains data on several variables including information on whether the firm exports or not and, if it exports, to which markets, total sales, employment, training activities for employees, input sourcing structure, investments aimed at improving already existing products and at introducing new products.

Estimation results suggest that sunk entry costs play an important role in shaping export decisions and that those firms with larger employment levels and average sales per employee are more likely to export. This is in line with findings reported in the existing empirical literature (see, e.g., Bernard and Jensen, 2004; Wagner, 2001; Arnold and Hussinger, 2005). Furthermore, we find that sourcing from abroad and investment in product improvement are associated with increased probability to export. After introducing these explanatory variables together, there is no residual heterogeneity and thus panel probit estimates are not significantly different from the

³ Das, Roberts and Tybout (2001) find that the option value of export market participation is quantitatively important for small scale exporters among the Colombian chemical producers, whose foreign demand is relatively limited.

⁴ MERCOSUR is a trade agreements established in 1991 by Argentina, Brazil, Uruguay, and Paraguay.

pooled ones. Finally, training activities for employees raise the odds of export outside of MERCOSUR.

The remainder of the paper is organized as follows. Section 2 reviews the theoretical and empirical literature on the determinants of export behavior. Section 3 describes the dataset and presents some descriptive evidence. Section 4 explains the empirical methodology. Section 5 discusses the estimation results. Section 6 concludes.

2 Export Behavior of Firms: Theory and Evidence

Exporting is a rather complex decision. Many factors are likely to affect whether a firm exports or only serves the domestic market. Over the last two decades, several contributions to the trade literature have highlighted the role played by various factors. These can be broadly classified into two groups: *sunk export costs* and forces affecting the profitability of exporting net of sunk costs, including firms' individual *characteristics*, *actions*, and *environment*.

Sunk costs are at the core of the theoretical models developed by Baldwin (1988), Baldwin and Krugman (1989) and Dixit (1989). These models highlight the interactions between the sunk costs associated with selling abroad and expectations formed in an uncertain environment. They predict that, due to sunk costs, current foreign market participation is affected by previous export experience. Specifically, firms that have sunk their entry costs and begun to export in response to a shock, such as a large devaluation, may not cease to export when the shock is reversed, so persistence in exporter status should be observed. By now several microeconometric studies have brought this prediction to data in the wake of Roberts and Tybout (1997). Their empirical approach is particularly designed to assess the role of the two aforementioned groups of factors as it takes into account that a firm's exports in a given year if the current and expected revenues are greater than the current period costs plus sunk entry costs, and the resulting net profit of exports hinges upon firm-specific attributes. The overall evidence is consistent with the theoretical prediction: sunk costs seem to be large and a significant source of export persistence (e.g., Roberts and Tybout, 1997; Bernard and Jensen, 1999, 2004; Bernard and Wagner, 2001; Arnold and Hussinger, 2005). After controlling for other sources of persistence, the probability that a firm will export can be more than 60% higher if it exported last period.⁵

Turning to firms' *characteristics*, further theoretical work by Melitz (2003) has shown that the existence of entry costs into the export market affects the distribution of the gains from trade

3

⁵ Importantly, evidence based on data for three manufacturing sectors in Colombia (basic chemicals, leather products, and knitted fabrics) suggest that for smaller firms the option value of being able to export in future years with incurring in entry costs is likely to substantially exceed the expected export profits in the current year (see Das et al., 2007).

across heterogeneous firms. In particular, only a portion of more efficient firms enter the export market and reap the benefits from trade in the form of larger market share and profit, whereas less efficient firms continue to produce only for the domestic market or disappear altogether.⁶ In other words, more efficient firms self-select into the export markets.⁷

From an empirical point of view, these theoretical results suggest that the heterogeneity of firm characteristics can play an important role in shaping export behavior. Size is one of these characteristics. As measured by the number of workers or the capital stock, it has been found to be positively related to export propensity (e.g., Roberts and Tybout, 1997; Hasan and Raturi, 2003; Bernard and Jensen, 2004). In this respect, size may account for diverse effects. On the one hand, it may serve as a proxy for productivity. Firms with lower average or marginal costs and thus higher efficiency tend to grow relative to the others. 10 On the other hand, size may capture scale-economy based exporting (e.g., Krugman, 1984). A second characteristic is age, which may be also considered as reflecting cost differences across producers. If market forces induce inefficient producers to exit, then older firms tend to be more competitive in world markets, either because of cost advantages that cannot be imitated by rivals or because they have had time to move down a learning curve. Even if the annual payoff from exporting were the same for young and old firms, the young ones would perceive smaller returns to entering the export markets since they have lower survival probabilities (see Tybout, 2001). A few studies provide empirical support to this effect (e.g., Roberts and Tybout, 1997 and Tucci, 2005). The structure of ownership (proprietorship and partnership vs. corporation or national vs. multinational) is a third characteristic that can also affect the odds of exporting (e.g., Brainard, 1997; Roberts and Tybout, 1997; Aitken et al., 1997; Greenaway et al., 2006). More precisely, multinationals are more likely to export as, in virtue of their multi-market presence, have better information about foreign markets and can exploit their already established distribution networks.

Firms' *actions* may also influence the profitability of exporting and thereby a firm exporting status. First, product quality may be a key factor for a firm to enter certain export markets (e.g.,

-

⁶ The exposure to trade only benefits the most efficient firms. The least efficient ones are forced to exit the industry. The implied trade-induced reallocations explain why trade may result in aggregate productivity gains without improving individual firms' productivity. For further details see Melitz (2003).

⁷ The underlying heterogeneity due to persistent differences across firms in terms of gross profit from exporting might also lead to persistence of exporter status (see Tybout, 2001).

⁸ Empirical evidence is abundant. However, the direction of causality is not always well established. In many cases we do not know whether firms with certain characteristics become exporters or firms get particular attributes by becoming exporters.

⁹ Caves (1989), Berry (1992), and Wagner (2001) review the evidence relating size and propensity to export.

¹⁰ Some papers introduce an explicit measure of productivity such as those estimated in the wake of Olley and Pakes (1996) or Levinsohn and Petrin (2003). The data consistently suggest a positive correlation between productivity at the firm level and exporting (e.g., Clerides et al., 1998; Bernard and Jensen, 1999; Kraay, 1999; Aw et al., 1997; Aw et al., 2000; Pavnick, 2002; Castellani, 2002; Delgado et al., 2002; Head and Ries, 2003; Hwang, 2003; Alessandria and Choi, 2007).

Brooks, 2006).¹¹ More specifically, firms may start exporting after improving or changing their products. Findings in a few papers lend support to this argument (e.g., Bernard and Jensen, 1999, 2004; Arnold and Hussinger, 2005). Second, the sourcing structure of inputs may also play an important role as long as inputs from different sources differ in terms of quality and incorporated technology (see, e.g., Hasan and Raturi, 2003; Tucci, 2005). If imported are better along these dimensions, they may increase the efficiency of the production process and, therefore, the probability of exporting. Similar effects may arise if input variety is richer in the foreign than in the domestic markets, thus allowing for a better match between the input mix and the production technology or the desired product characteristics (see Either, 1982).. Furthermore, having contact with a foreign supplier may help entry into export markets by attenuating informational problems. Lastly, in the case of firms that participate in international production networks, inputs requirements may be part of licensing agreements so that firms must first import specific inputs from contractually defined sources before they can process and export their products.

Finally, the *environment* in which firms operate can also affect their export behavior. First, in the presence of significant differences across regions in terms of infrastructural quality and access to foreign markets, firm location may be a critical determinant for the feasibility of exporting (see, e.g., Roberts and Tybout, 1997). Second, different sectors of activity may be associated with different probabilities of exporting since not only the strength of a country's comparative advantage but also intra-industry patterns of producer heterogeneity and international demand vary across sector (see, e.g., Roberts and Tybout, 1997; and Das et al., 2007). Third, firms may benefit from spillovers from other firms. In particular, proximity to exporters, both from the technological and the geographical points of view, may lead to informational spillovers, lower costs of access to foreign markets and, thereby, higher probabilities of exporting. The evidence on these effects is mixed as existing export activities have been found to have positive, zero, or negative impacts on other firms' exports in the same sector or region (see, e.g., Aitken et al., 1997; and Bernard and Jensen, 2004).

All these findings are generally based on firms sampled across all possible size segments. This raises two interesting questions when it comes to SMEs: Do those findings also hold for the lower tail of the firm size distribution? Are the factors that appear to be important in general also relevant in the specific case of the export behavior of SMEs? In the next sections, we will provide answers to these questions by exploring how sunk costs as well as firms' characteristics, actions, and environment have affected SMEs' export outcomes in a developing country, namely

_

¹¹ If high product quality requires high workforce quality, a positive link between the level of qualification of the employees and the probability to export may be expected. Evidence on the impact of this variable, as proxied by average

Argentina.

3 Data and Descriptive Evidence

We use data from the "Observatorio PyME" database collected by the University of Bologna's Representation in Argentina for the years 1996-1998. The universe consists of all industrial firms that according to the 1994 National Economic Census, have between 10 and 200 employees and annual revenues up to 20 millions pesos.12 The size of this universe is 9,683 firms that jointly employed 305,425 workers (i.e., 30% of total manufacturing employment in 1994). From this universe, firms were sampled using panel techniques. In particular, the sample was designed following a three step procedure proposed by the INDEC (Argentina's Statistical Bureau). First, employment level strata were identified such that their limits were defined according to Neumann's optimal allocation method when the same number of cases in each stratus is assumed: Stratus 1: 10-24 employees; Stratus 2: 25-54 employees; and Stratus 3: 55-100 employees. Second, six regions were defined to study the sample size distribution for each of them that corresponds to an optimal allocation. This distribution was used to determine the sample size for each region.¹³ Third, within each region, this sample was distributed across the employment strata according to the optimal allocation method taking production value as the reference value. It should be stressed that the coefficients of variation of the estimations derived from simple expansions of production values are statistically acceptable, i.e., estimations are precise (expected error below 7% in all cases). Hence, the sample is statistically representative.

192 one-plant firms completed all three surveys carried out by "Observatorio PyME" each year over the period 1996-1998. This is the sample we use in our empirical analysis. The database provides basic information such as location and main activity (i.e., industry the firm belongs to), and includes data on sales, employment, date of creation, membership of an economic group, training, investments to introduce a new product, investments to improve an existing product, main origin of inputs used in the production process, exports and their main destinations. The exact definition of these variables can be found in the Appendix.

Table 1 reports the number of exporting firms over the sample period and the average degree

wages and the ratio of white collars to total employees, is however not robust (see Roberts and Tybout, 1997; Bernard and Jensen, 1999, 2004; Bernard and Wagner, 2001; Arnold and Hussinger, 2005).

¹² In order to ensure comparability of firms' economic and financial data coming from their balance sheets, additional exclusion criteria have been used. Thus, the data set also excludes firms that due to their legal status are difficult to survey (i.e., one-person and de facto firms), those that are not relevant for the purpose of the survey (i.e., state-owned firms and non-profit institutions), and firms that are not obliged to submit balance sheets.

¹³The six regions consist of the following provinces. Buenos Aires: Province of Buenos Aires; Centro: City of Buenos Aires, Santa Fe, and Córdoba; Cuyo: San Juan, San Luis, and Mendoza; NEA: Formosa, Chaco, Misiones, Entre Ríos, and

of export intensity, i.e., the ratio of exports to total sales, whereas Table 2 shows the distribution of firms across regions and exporter status for each sample year. The number of exporting firms has decreased between 1996 and 1998 and the same is true for the average export share. This drop coincides with both a real exchange appreciation of the Argentine peso (see, e.g., Escudé et al., 2001) and a worsening of macroeconomic conditions in the Southern Cone provoked by the Asian and Russian crises, especially in Brazil (see e.g., Werneck, 2009), which was the main destination of manufacturing exports. Firms are primarily located in two regions: Buenos Aires and Centro. This is far from surprising. These two regions accounted for almost 80% of total manufacturing employment in 1994. No significant differences are observed in the share of exporting firms across most regions, except for the region Sur, which systematically exhibits lower shares.

Transition rates into and out of the export markets are reported in Table 3. Though there is clearly substantial persistence in the export status of individual plants, these transition rates indicate that year-to-year transitions for Argentinean SMEs are relatively large. Thus, 27.10% of the firms in the export market in 1996 exited the market in 1997, while 14.81% of firms that did not export in 1997, started to export in 1998.¹⁴

Table 4 presents the mean of the main variables for exporting firms and non-exporting firms along with their difference. Exporters are larger (in terms of employment), have larger sales per employee, are more active in training activities, invest more in product improvement, and belong to an economic group. Similar conclusions can be reached by looking at the "exporter premia", i.e., the average percentage differences between exporters and non-exporters in the same industry, state, and year (see Bernard and Jensen, 1999) shown in Table 5. Note that, in this case, estimates suggest that exporters mainly use inputs from abroad.

4 Empirical Methodology

As discussed in Section 2, export behavior can be explained in terms of sunk costs as well as firms' characteristics, actions, and environment. In this section, we capture these determinants through the export market participation equation and describe the empirical methodology used to estimate it. In particular, the decision to export by the firm is modeled following Roberts and

Corrientes; NOA: Jujuy, Salta, Tucumá n, Catamarca, La Rioja, and Santiago del Estero; Sur: La Pampa, Rí o Negro, Neuqué n, Santa Cruz, and Tierra del Fuego.

¹⁴ Sánchez et al. (2008) examine the determinants of the emergence of three new successful activities in Argentina, blueberries, chocolate confection, and biotechnology applied to human health. In particular, they analyze who were the pioneering companies, what were the *ex ante* uncertainties regarding the profitability of exports, how these uncertainties were overcome, and to what extent business opportunities diffused. In this analysis, they consider the role played by previously accumulated capabilities, industry-specific public goods, and public policies.

Tybout (1997) and Bernard and Jensen (1999, 2004).¹⁵ Formally, in the one period case and in the absence of entry costs, the firm maximizes current profits:

$$\pi_{ii}(q_{ii}^*, Z_{ii}, Y_{ii}) = p_{ii}q_{ii}^* - c_{ii}(q_{ii}^*, Z_{ii}, Y_{ii})$$
(1)

where p_{it} is the export price of goods sold abroad by the firm i in period t; c_{it} is the variable cost of producing the profit-maximizing level of exports q^*_{it} ; Z_{it} is a vector of firm-specific features that influence the probability of exporting and include both firms' characteristics and actions such as size, age, sourcing structure, etc; Y_{it} is a vector of environmental factors that are exogenous to the firm and affect its probability of exporting such as industry demand shocks and region-industry spillovers. A firm exports if expected profits are greater than zero, i.e. if current revenues are larger than current period costs:

$$X_{ii} = \begin{cases} 1 & if \quad \pi_{ii}(q_{ii}^*, Z_{ii}, Y_{ii}) = p_{ii}q_{ii}^* - c_{ii}(q_{ii}^*, Z_{ii}, Y_{ii}) \rangle \ 0 \\ 0 & otherwise \end{cases}$$
 (2)

where X_{it} is binary variable indicating the export status of the firm i in period t.

In a multi-period setting and in the presence of entry costs, the firm chooses a sequence of output levels in order to maximize current and discounted future profits:

$$\Pi_{it} = E_t \left\{ \sum_{s=t}^{\infty} \delta^{s-t} \left[\widetilde{\pi}_{is} \left(q_{is}^*, Z_{is}, Y_{is} \right) X_{is} \right] \right\} = E_t \left\{ \sum_{s=t}^{\infty} \delta^{s-t} \left[p_{ts} q_{is}^* - c_{is} \left(q_{is}^*, Z_{is}, Y_{is} \right) - N(1 - X_{is-1}) \right] X_{is} \right] \right\}$$
(3)

where N is the entry cost for the firm. These costs may include the costs of gathering information on demand conditions in foreign markets and those of establishing a distribution system that can be thought as sunk in nature. The firm does not pay these costs if it exported in the previous period, i.e., X_{it-1} =1. The existence of sunk entry costs makes the decision rule dynamic because exporting today implies an additional option value of being able to export tomorrow without facing again those costs. In other words, revenues in each period t are equal to revenues from export sales in that period plus any discounted increase in the value of the firm from exporting in t. Formally, the value function for the dynamic programming problem is given by:

$$V_{ii} = \max_{X_{ii} \in [0,1]} \{ \tilde{\pi}_{ii} [X_{ii} = I] \} + \delta E_t [V_{ii+1}(.) | X_{ii}]$$
(4)

The solution to this problem is the following decision rule:

$$X_{ii} = \begin{cases} 1 & \text{if} \quad p_{ii}q_{ii}^* + \delta\{E_t[V_{ii+1}(.)|X_{ii} = 1] - [V_{ii+1}(.)|X_{ii} = 0]\} - [c_{ii}(q_{ii}^*, Z_{ii}, Y_{ii}) + N_{ii}(1 - X_{ii-1})] \rangle 0 \\ 0 & \text{otherwise} \end{cases}$$
(5)

The firm therefore exports if expected profits are greater than zero in present value, i.e., if current and expected future revenues are larger than the current period costs c_{it} plus any costs of entry. Accordingly, the actual decision to export in a particular period depends on whether the

¹⁵ This is a reduced-form model. Das et al. (2007) develop a dynamic structural model featuring uncertainty, plant-level heterogeneity in export profits, and sunk entry costs for firms breaking into foreign markets that characterizes firms' decisions about whether to export as well as quantifies the volume of foreign sales among those who do.

firm has exported in the previous period (i.e., lagged exporter status), firm-specific features Z_{it} and factors exogenous to the firm Y_{it} . The theoretical decision rule can then be expressed as an empirical binary choice model of the form:

$$X_{ii} = \begin{cases} 1 & if \quad \beta Z_{ii} + \gamma Y_{ii} - N(1 - X_{ii-1}) + \varepsilon_{ii} > 0 \\ 0 & otherwise \end{cases}$$
 (6)

Operationally, we quantify the effect of the aforementioned factors on the probability of exporting by estimating the following probit model:

$$P(X_{it} = 1) = \Phi(\alpha + \lambda X_{it-1} + \beta' Z_{it-1} + \gamma' Y_{it-1} + \tau_t + \varepsilon_{it})$$
(7)

where Z_{it-1} ={employment_{it-1}, age_{it-1}, training_{it-1}, group_{it-1}, domestic inputs_{it-1}, product improvement_{it-1}, product introduction_{it-1}, sales per employee_{it-1}}; Y_{it-1} ={industry fixed effects, region fixed effects, exporting neighbors_{it-1}}; and τ_t denotes year fixed effects that control for the role of macroeconomic factors such as business cycles and real exchange rate (see, e.g., Alessandria and Choi, 2007; and Blanes-Cristóbal et al., 2008). The main variables are defined in Table 6 and broadly correspond to those factors whose general influence on firms' export behavior has been reviewed in Section 2. Notice that these explanatory variables are lagged one period to avoid simultaneity problems. Moreover, since we need to include a lagged dependent variable, estimations will be performed over the period 1997-1998.

As highlighted by Bernard and Jensen (1999, 2004), there likely exist unobserved characteristics, such as product attributes or managerial ability, affecting the decision to export. These characteristics are potentially permanent so they will lead to persistence in export behavior and thus failing to account for them can result in the overestimation of the entry costs. These unobserved firm characteristics can be formally modeled assuming that the error term has two components: a permanent firm-specific component v_i and the remainder (transitory)

¹⁶ Using a general equilibrium model where firms face an up-front sunk cost of entering foreign markets and smaller period-by-period continuation cost, Alessandria and Choi (2007) show that business cycles affect the moments when firms start or stop exporting. In particular, during economic expansions, more firms start exporting and more exporters are likely to stay in export markets than in normal times.

¹⁷ In general, employment and age are defined as in most papers in the existing empirical literature. The role of imported (as opposite to domestic) inputs has been captured by Tucci (2005) through the share of expenditures on imported intermediate inputs in the total expenditures on intermediate inputs. Product improvements and product introduction can be considered to be (at least partially) outputs of innovation activities, whose influence in shaping firms' foreign market participation has been accounted for by, among others, Arnold and Hussinger (2005) and Blanes-Cristóbal et al., 2008) using measures involving expenditure in R&D. Arnold and Hussinger (2005) also explicitly control for the impact of new products by including the share of new products in firms' total sales, whereas Bernard and Jensen (2004) incorporate a binary variable that identifies whether firms report to different industries over time. Spillovers can take place at different levels: region, industry, and region-industry. In the latter case, Bernard and Jensen (2004) proxy these effects using the share of exporting plants (exports) in total plants (shipments) in the same region and the same industry (excluding the plant in question).

¹⁸ We report the pair-wise correlation between these variables in Table A1 in Appendix A. The correlations clearly indicate that multicollinearity does not seem to be an issue in our estimations.

disturbance μ_{ii} . Introducing this feature, we get a dynamic binary panel data model. We will estimate this model as a probit with random effects:¹⁹

$$P(X_{it} = 1) = \Phi(\alpha + \lambda X_{it-1} + \beta' Z_{it-1} + \gamma' Y_{it-1} + \nu_i + \tau_t + \mu_{it})$$
(8)

Firms not only choose whether to export or not at all, but also whether to export or not to a particular market. Our database provides information on export presence in two main destination markets, MERCOSUR and the rest of the world, and thus allows us to examine whether the relative importance of the determinants of exports decisions varies across these markets. We believe that distinguishing between these destinations is interesting and worthy of being explored. In general, as discussed above, factors influencing export behavior may differ across trade partners. More specifically, exporting to MERCOSUR implies exporting under tariff preferences, to geographically close and thereby more familiar countries, and probably to less demanding markets in terms of product quality and technological sophistication. Clearly, many of these conditions do not necessarily hold when exporting to the rest of the world. Being active therein would seem a priori more difficult. Hence, firms' characteristics and actions required to break into countries of this group can be expected to differ from those relevant to export to MERCOSUR partners.

We therefore extend our original probit model in a way that resembles the seemingly unrelated regressions model, i.e., we specify one equation for each market (MERC for MERCOSUR and ROW for rest the world) and assume that the disturbances are correlated:

$$P(X_{it}^{MERC} = 1) = \Phi(\alpha^{MERC} + \lambda^{MERC} X_{it-1} + \beta^{MERC'} Z_{it-1} + \gamma^{MERC'} Y_{it-1} + \tau_t^{MERC} + \varepsilon_{it}^{MERC})$$

$$(9)$$

$$P(X_{it}^{ROW} = 1) = \Phi(\alpha^{ROW} + \lambda^{ROW} X_{it-1} + \beta^{ROW} Z_{it-1} + \gamma^{ROW} Y_{it-1} + \tau_t^{ROW} + \varepsilon_{it}^{ROW})$$

$$with \quad Cov(\varepsilon_{it}^{MERC}, \varepsilon_{it}^{ROW}) = \rho$$

$$(10)$$

This model can be estimated by maximum-likelihood and the null hypothesis that the correlation of the error terms is equal to zero can be tested with a Wald test (Greene, 1997). We now turn to the estimation results.

¹⁹ Ensuring consistency in this framework requires solving the initial conditions problem. In linear models with additive unobserved effects, this problem can be solved using a transformation such as differentiation to eliminate these effects and then applying instrumental variables to perform generalized method of moments (GMM) estimations. In the

nonlinear case, solving the problem is much more difficult. Woodridge (2005) proposes, however, a simple solution consisting of modeling the distribution of the unobserved effect conditional on the initial value and any exogenous variable. Operatively, this implies including as explanatory variables the exporter status in the initial year, the exporter status in the previous year, and a vector of all (non-redundant) explanatory variables in all time periods to allow for correlation between these regressors and the unobserved effect in all years. Given the short extension of our panel, doing this creates severe multicollinearity problems, which unfortunately prevent us from using this strategy. The impossibility

5 Estimation Results

Table 7 reports Probit estimates of Equation (7). Region-, industry-, and year-fixed effects are included but not reported. Sunk entry costs seem to be an important determinant of export decisions by SMEs in Argentina. Lagged exporter status has a large and significant impact on current export activity. Thus, having been an exporter the period before is associated with higher probability of being an exporter in the present period. In particular, a discrete change from zero to one in the variable representing lagged exporter status in the preceding year increases the probability of exporting by almost 70%, at the mean of all remaining variables (see Table A2 in the Appendix). Moreover, the estimated coefficient on employment is positive and significantly different from zero suggesting that large firms are more likely to export. These results coincide with those reported in the existing empirical literature (e.g., Roberts and Tybout, 1997; Bernard and Jensen, 1999, 2004; and Arnold and Hussinger, 2005).

Sourcing patterns are also relevant to understanding the export behavior of SMEs in Argentina. More specifically, firms that rely more heavily on domestic providers are less likely to export. This is in line with findings in Hasan and Raturi (2003) and Tucci (2005).

Improving products seems to pay off in terms of export performance. Investing in enhancements of already commercialized goods increases the probability of exporting by approximately 20% as compared with no investment, when all remaining variables are their mean values (see Table A2 in the Appendix).

Several papers assess the effect of productivity on firm's export behavior (see, e.g., Bernard and Jensen, 2004; Arnold and Hussinger, 2005). Unfortunately, we are unable to estimate a measure of total factor productivity following the procedures proposed by Olley and Pakes (1996) or Levinsohn and Petrin (2000) because we do not have the required data. To control for this factor and in addition to employment and age, we include total sales divided by employment as raw proxy for average productivity. Of course, this variable is subject to many caveats and these should be kept in mind when interpreting the corresponding estimates. We find that average sales per employee are positively and significantly associated with the probability of exporting.

Training, product introduction, and closeness to exporters in the same industry and region display the expected sign but are not significantly different from zero. This may be related to the temporal dimension of their effects. Thus, to substantially change their export behavior, firms may need to learn for more than one year about new products launched to the market or from

specific effect should be kept in mind when considering the probit random effect estimates reported below. Recall that the

other exporters and to develop training activities over longer periods in order to alter their production process or the quality of their products. The remaining variables, i.e., age and belonging to larger economic group are not significant either.²⁰

Including a lagged dependent variable and using one-year lagged regressors imply that our estimations only consider firms that participated in all surveys. However, there are firms that did not answer the questionnaires as well as firms that only answered some of them. It is clear that there may be systematic differences between the firms that always answered and the firms that did not answer all the questionnaires. In short, there may be a sample selection bias, which could potentially make our estimates inconsistent. In order to address this issue, we have reconstructed the universe by industry and region using the weighting factors in our sample. This estimated universe has then been utilized to post-stratify the firms in our panel and accordingly generate a new weighting factor. Finally, we have used this new factor to re-estimate our probit models, thereby accounting for the probability that the firms appear in the sample and therefore controlling for potential selection biases created by our economic approach. In doing this, we explicitly consider the strata structure of our data.²¹ Estimations results are shown in Table 8.²² These results coincide with those reported above.

Table 9 presents estimation results when standard errors are bootstrapped. Note that these errors are now larger. However, the main message remains the same: size, sourcing intermediate inputs from abroad, investing in product improvement and average sales per employee are all associated with larger export probabilities. Table 10 shows estimates obtained when the sample is restricted to firms that do not change their status in terms of export participation ("non-switchers"). This somehow arbitrary subsample has a valuable property, namely, it allows us to abstract from the effect of lagged exporter status and thus, to check for the robustness of the results for the remaining variables (see Arnold and Hussinger, 2005). Most results are qualitatively the same as before. Note that performing training activities for employees significantly now increases the odds of exports.

As mentioned above, there might be significant unobserved plant effects and ignoring these effects when present would lead to biased estimates. To account for this unobserved heterogeneity, we estimate Equation (8) using a random-effects probit estimator. Results are reported in Table 11 and are essentially the same as those shown before. More precisely, as

probit model does not lend itself to the fixed effect treatment (see Greene, 1997).

²⁰In additional estimations, we also incorporate leverage as an additional explanatory variable (see Greenaway et al., 2006). This variable turns out to be non-significant. These results are not shown but are available from the authors upon request.

²¹ These estimates are obtained assuming stratification by region. Similar results are found when stratification by industry is assumed instead. These results are not reported but are available from the authors upon request.

²² Estimations have been performed using STATA's command for survey estimation svy.

suggested by the likelihood ratio test, the panel variance component is unimportant and hence the panel estimator is not significantly different from the pooled probit estimator.

We now turn to export behavior across two main destination markets, MERCOSUR and the rest of the world. Table 12 presents seemingly unrelated biprobit regressions. Note first that, according to the Wald test, the null hypothesis that the correlation of the disturbances is equal to zero should be rejected. Probit equations in the model are therefore not independent and hence cannot be estimated separately. Sunk entry costs are similarly important for exports to both markets. Firms exporting to MERCOSUR essentially have the same characteristics identified previously: they are larger, mainly sourced from abroad, have invested in improving their products, and have larger average sales per employee. For exports to the rest of the world, training activities seem to be a key factor.

6 Concluding Remarks

A sizable body of research has recently focused on the export behavior patterns of firms. Most empirical investigations concentrate on developed countries and, by construction of the corresponding datasets, on relatively larger firms. Evidence on smaller firms in developing countries is therefore scarce. This paper has attempted to fill this gap in the literature by analyzing the export decisions of a statistically representative sample of 192 SMEs in Argentina over the period 1996-1998. In so doing, we have used a new database created by the University of Bologna's Representation in Argentina. This database contains information on standard key variables such as size, age, sales, exports, training activities, inputs sourcing patterns, investment in introducing and improving already existing products. It also provides information on the main destination of exports (MERCOSUR vs. rest of the world). Admittedly, although rich, these data have two main limitations that should be kept in mind when interpreting the estimation results: first, the sample period is short and, second, most explanatory variables are binary.

Our pooled probit estimates suggest that sunk entry costs play a significant role in determining export decisions. Furthermore, larger firms and those with greater average sales per employee are more likely to export. Sourcing from abroad and investment in product improvement are also associated with an increased probability of exporting. After introducing these explanatory variables together, there is no residual heterogeneity and thus random-effects panel probit estimates are not significantly different from the pooled ones. Finally, according to our seemingly unrelated bivariate estimates, the training of employees helps raise the odds of exporting outside of MERCOSUR.

References

- Aitken, B., G. Hanson, and A. Harrison, 1997. Spillovers, foreign investment, and export behavior. Journal of International Economics, 43, 1-2.
- Alessandria, G. and H. Choi, 2007. Do sunk costs of exporting matter for net export dynamics. Quarterly Journal of Economics, 122, 1.
- Álvarez, R., 2004. Sources of export success of small and medium-sized enterprises: The impact of public programs. International Business Review, 13.
- Anderson, J., 2000. Why do countries trade (so little)? Pacific Economic Review, 5.
- Arnold, J. and K. Hussinger, 2005. Export behavior and firm productivity in German manufacturing: A firm level analysis. Weltwirtschaftliches Archiv, 141, 2.
- Aw, B., X. Chen, and M. Roberts, 1997. Firm-level evidence on productivity differentials, turnover, and exports in Taiwanese manufacturing. NBER Working Paper 6235.
- Aw, B., S. Chung, and M. Roberts, 2000. Productivity and turnover in the export market: Micro evidence from Taiwan and South Korea. World Bank Economic Review, 14, 1.
- Baldwin, R., 1988. Hysteresis and the beachhead effects. American Economic Review, 78.
- Baldwin, R. and P. Krugman, 1989. Persistent trade effects of large exchange rate shocks. Quarterly Journal of Economics, 104.
- Barrios, S.; Görg, H.; and Strobl, E., 2003. Explaining firms' export behaviour: R&D, spillovers, and the destination market. Oxford Bulletin of Economics and Statistics, 65, 4.
- Bernard, A. and B. Jensen, 1999. Exceptional exporter performance: Cause, effect, or both? Journal of International Economics 47, 1.
- Bernard, A. and B. Jensen, 2004. Why some firms export. Review of Economics and Statistics, 86, 2.
- Bernard, A. and J. Wagner, 2001. Export entry and exit by German firms. Weltwirtschaftliches Archiv, 137, 1.
- Berry, R., 1992. Firm (or plant) size in the analysis of trade and development, in G. Helleiner, ed., Trade policy, industrialization, and development: New perspectives. Oxford: Clarendon Press.
- Blanes-Cristobal, J, M. Dovis, J. Milgram-Baleix, and A. Moro-Egido, 2008. Do sunk exporting costs differ among markets? Evidence from Spanish manufacturing firms. Economic Letters, 101, 2.
- Brainard, L., 1997. An empirical assessment of the proximity/concentration trade-off between multinational sales and trade. American Economic Review, 87, 4.
- Brooks, E., 2006. Why don't firms export more? Product quality and Colombian plants. Journal of Development Economics, 80, 1.
- Castellani, D., 2002. Export behavior and productivity growth: Evidence from Italian manufacturing firms. Weltwirtschaftliches Archiv, 138, 4.
- Castellani, D., F. Serti, and C. Tomasi, 2009. Firms in international trade: Importers and exporters heterogeneity in the Italian manufacturing industry. The World Economy, forthcoming
- Caves, R., International differences in industrial organization, in R. Schmalensee and R. Willig, eds., Handbook of Industrial Organization, 2. Amsterdan: North-Holland.
- Clerides, S., S. Lach, and J. Tybout, 1998. Is learning by exporting important? Micro-dynamic evidence from Colombia, Mexico, and Morocco. Quarterly Journal of Economics.
- Das, S.; M. Robert; and J. Tybout, 2007. Market entry costs, producer heterogeneity, and export dynamics. Econometrica, 75, 3.
- Delgado, M., J. Fariñas, and S. Ruano, 2002. Firm productivity and export markets: A nonparametric approach. Journal of International Economics, 57, 2.
- Dixit, A., 1989. Entry and exit decisions under uncertainty. Journal of Political Economy, 97.
- Eaton, J., S. Kortum, and F. Kramarz, 2004. Dissecting trade: Firms, industries, and export destinations. American Economic Review, 94, 2.
- Eaton, J., S. Kortum, and F. Kramarz, 2008. An anatomy of international trade: Evidence from

- French firms. Mimeo.
- Either, W., 1982. National and international returns to scale in the modern theory of international trade. American Economic Review, 72, 3.
- Escudé, G., M. Gabrielli, and V. Cohen Saban, 2001. Evolución del tipo de cambio real multilateral de Argentina en los últimos 10 años. Nota Técnica 11, Gerencia de Investigaciones Económica-Financieras, Banco Central de la República Argentina.
- Girma, S., D. Greenaway, and R. Kneller, 2004. Does exporting increase productivity? A microeconometric analysis of matched firms. Review of International Economics, 12, 5.
- Greenaway, D., A. Guariglia, R. Kneller, 2007. Financial factors and exporting decisions. Journal of International Economics, 73, 2.
- Greene, W., 1997. Econometric Analysis. Prentice Hall, New Jersey.
- Grossman, G., 1998. Comments on Alan V. Deardorff, Determinants of bilateral trade: Does gravity work in a neoclassical world?, in J. Frankel, ed., The regionalization of the world economy. University of Chicago for NBER, Chicago.
- Guiso, L., P. Sapienza, and L. Zingales, 2008. Cultural biases in economic exchange. Quarterly Journal of Economics, forthcoming.
- Hallak, J., 2006. Product quality and the direction of trade. Journal of International Economics, 68,
- Hasan, R. and M. Raturi, 2003. Does investing in technology affect exports? Evidence from Indian firms. Review of Development Economics, 7, 2.
- Head, K. and J. Ries, 2003. Heterogeneity and the FDI versus export decision of Japanese Manufacturers. Journal of the Japanese and International Economies, 17, 4.
- Hwang, A., 2003. Exports, returns to scale, and total factor productivity: The Case of Taiwanese Manufacturing Industries. Review of Development Economics, 7, 2.
- Hwang, R., 2007. Distance and trade: Disentangling unfamiliarity effects and transport cost effects. European Economic Review, 51, 1.
- Kneller, R., M. Pisu, and Z. Yu, 2008. Overseas business costs and firm export performance. Canadian Journal of Economics, 41, 2.
- Kraay, A., 1999. Exportations et performances economiques: Etude d'un panel d'entreprises Chinoises. Revue d'Economie du Developpement, 7, 1-2.
- Krugman, P., 1984. Import protection as export promotion: International competition in the presence of oligopoly and economies of scale, in H. Kierzkowski, ed., Monopolistic competition and international trade. Oxford: Oxford University Press.
- Lawless, M., 2009. Firm export dynamics and the geography of trade. Journal of International Economics, Elsevier 77, 2.
- Levinsohn, J. and A. Petrin, 2003. Estimating production functions using inputs to control for unobservables. Review of Economic Studies, 70.
- Loungani, P., A. Mody, and A. Razin, 2002. The global disconnect: The role of transactional distance and scale economies in the gravity equation. Scottish Journal of Political Economy, 49.
- Melitz, M., 2003. The impact of trade on intra-industry reallocations and aggregate industry productivity. Econometrica, 71, 6.
- Melitz, M. and G.I.P. Ottaviano, 2008. Market size, trade, and productivity. Review of Economic Studies, 75, 1.
- Olley, S. and A. Pakes, 1996. The dynamics of productivity in the telecommunications equipment industry. Econometrica, 64, 6.
- Pavnick., N., 2002. Trade liberalization, exit, and productivity improvements: Evidence from Chilean plants. Review of Economic Studies, 69, 1.
- Portes, R., H. Rey, and Y. Oh, 2001. Information and capital flows: The determinants of transactions in financial assets. European Economic Review, 45, 4-6.
- Roberts, M and J. Tybout, 1997. The decision to export in Colombia: An empirical model of entry with sunk costs. American Economic Review, 87, 4.

- Sánchez, G, R. Rozemberg, I. Butler, and H. Ruffo, 2008. The emergence of new successful export activities in Argentina: Self-discovery, knowledge niches, or barriers to riches? Research Network Working Paper 548, Inter-American Development Bank.
- Tucci, A., 2005. Trade, foreign networks, and performance: A firm-level analysis for India. London School of Economics, mimeo.
- Tybout, J., 2001. Plant- and firm-level evidence on "new" trade theories. NBER Working Papers 8418.
- Wagner, J., 2001. A note on the firm-size export relationship. Small Business Economics, 17.
- Werneck, R., 2009. Setting up a modern macroeconomic policy framework in Brazil, 1993-2004. Commission on Growth and Development, Working Paper 45.
- Wooldridge, J., 2005. Simple solutions to the initial conditions problem for dynamic, nonlinear panel data models with unobserved heterogeneity. Journal of Applied Econometrics, 20.

Table 1

Number of Exporting Firms and Average Export to Sales Ratio									
Year Exporting Firms Export Share (%)									
1996	107	27.751							
1997	84	15.153							
1998	89	13.661							

Table 2

Distribution of	Distribution of Firms across Regions and across Exporter Status by Region (%)									
Region	Total Number of	E	xporting Firm	าร						
Kegion	Firms	1996	1997	1998						
Buenos Aires	44.792	60.465	41.860	45.349						
Centro	40.625	51.282	43.590	47.436						
Cuyo	5.208	60.000	50.000	50.000						
NEA	2.604	80.000	80.000	60.000						
NOA	2.604	40.000	40.000	40.000						
Sur	4.167	37.500	37.500	37.500						

Table 3

Transitions In an	d Out of Export Marke	ts							
1	996-1997								
Status t\ Status t+1 Non-Exporter Exporter									
Non-Exporter	92.940	7.600							
Exporter	27.100	72.900							
1	997-1998								
Status t\ Status t+1	Non-Exporter	Exporter							
Non-Exporter	85.190	14.810							
Exporter	13.100	86.900							

Table 4

E	xporter vs. Non-Expor	rters	
Variable	Exporters	Non-Exporters	Difference
Employment	68.982	42.976	26.006***
Age	30.948	31.479	-0.531
Training	0.671	0.398	0.273***
Domestic Inputs	0.514	0.550	-0.036
Group	0.200	0.125	0.075**
Product Improvement	0.361	0.257	0.104**
Product Introduction	0.386	0.422	-0.036
Exporting Neighbors	0.769	0.697	0.072
Sales per Employee	121.733	63.733	58.000**

The table reports the average values of the variables for each exporter status as well as their differences. The significance of these differences is tested using a t-test for differences in mean. *** significant at 1% level; ** significant at 5% level; * significant at 10% level.

Table 5

Exporter vs. Non-l	Exporters
Variable	Exporter Premia
Employment	56.127*
Age	-3.275
Training	26.667*
Domestic Inputs	-14.448*
Group	7.983*
Product Improvement	8.452
Product Introduction	-8.076
Exporting Neighbors	0.039
Sales per Employee	30.356***

Sales per Employee 30.356***
The table reports the slope coefficient of an OLS (Probit) of each of the (binary) variables on a dummy characterizing the exporter status of the firm, industry-fixed effects, region-fixed effects, and year-fixed effects (multiplied by 100). * significant at least at 10% level.

Table 6

Variable	Definition
Export	Binary variable taking the value of 1 if the firm exports and 0 otherwise
Employment	Number of employees of the firm
Age	Difference between sample year and firm's creation year
Training	Binary variable taking the value of 1 if the firm has performed training activities for its employees the previous year and 0 otherwise
Domestic Inputs	Binary variable taking the value of 1 if the firm sources its inputs mainly from domestic providers (i.e., over 50%) and 0 otherwise
Group	Binary variable taking the value of 1 if the firm belongs to larger economic group and 0 otherwise
Product Improvement	Binary variable taking the value of 1 if the firm has invested in changing and improving the quality of its already existent products the previous year and 0 otherwise
Production Introduction	Binary variable taking the value of 1 if the firm has invested in introducing new products into the market the previous year and 0 otherwise
Exporting Neighbors	Binary variable taking the value of 1 if there has been at least one exporting firm in the same industry in the same region the previous year and 0 otherwise
Sales per Employee	Total sales divided by number of employees

Table 7

			Export D	ecision: Pro	obit Estima	tes				
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)
Export(-1)	2.132***	1.942***	1.936***	1.917***	1.936***	1.936***	2.118***	2.105***	2.118***	2.059***
	(0.204)	(0.208)	(0.209)	(0.210)	(0.216)	(0.216)	(0.231)	(0.231)	(0.232)	(0.235)
Employment		0.511***	0.528***	0.511***	0.516***	0.538***	0.716***	0.718***	0.715***	0.828***
		(0.124)	(0.131)	(0.133)	(0.136)	(0.141)	(0.168)	(0.170)	(0.170)	(0.169)
Age			-0.089	-0.079	-0.059	-0.056	-0.073	-0.065	-0.064	-0.074
			(0.107)	(0.107)	(0.108)	(0.108)	(0.119)	(0.122)	(0.122)	(0.128)
Training				0.114	0.149	0.156	0.138	0.152	0.155	0.036
				(0.183)	(0.187)	(0.189)	(0.208)	(0.215)	(0.216)	(0.224)
Domestic Inputs					-0.597**	-0.582*	-0.895***	-0.890***	-0.934***	-0.801**
_					(0.300)	(0.300)	(0.312)	(0.310)	(0.311)	(0.314)
Group						-0.163	-0.131	-0.146	-0.096	-0.259
						(0.223)	(0.235)	(0.240)	(0.253)	(0.272)
Product Improvement							0.450**	0.452***	0.456***	0.532***
							(0.177)	(0.176)	(0.175)	(0.168)
Product Introduction								-0.118	-0.109	-0.072
The second second NT of a factor and								(0.218)	(0.219)	(0.220)
Exporting Neighbors									0.266	0.362
Calas man Emmlayea									(0.295)	(0.293) 0.471***
Sales per Employee										(0.146)
Industry Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Industry Fixed Effects Region Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Year Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Pseudo-R ²	0.400	0.421	0.422	0.423	0.433	0.434	0.477	0.477	0.478	0.497
Observations The table reports probit estimate	376	367	367	367	367	367	330	330	330	328

The table reports probit estimates of Equation (7) for the period 1997-1998. Export is a binary variable taking the value of 1 if the firm exports and 0 otherwise (Export(-1) is the one-year lagged value). Employment is the natural logarithm of the firm employment level (lagged one year). Age is the natural logarithm of the firm's age (lagged one year). Training is a binary variable taking the value of 1 if the firm has performed training activities for its employees the previous year and 0 otherwise. Domestic Inputs is a binary variable taking the value of 1 if the firm sources its inputs mainly from domestic providers, i.e., over 50% and 0 otherwise (lagged one year). Group is a binary variable taking the value of 1 if the firm has invested in changing and improving the quality of its already existent product the previous year and 0 otherwise. Product Introduction is a binary variable taking the value of 1 if the firm has invested in introducing new products into the market in the previous year and 0 otherwise. Exporting Neighbors is a binary variable taking the value of 1 if there has been at least one exporting firm in the same industry in the same province the previous year and 0 otherwise. Sales per Employee is the natural logarithm of the firm's total sales divided by the number of employees (lagged one year). Robust standard errors clustered on firm are reported below estimated coefficients between parentheses. *** significant at 1% level; ** significant at 5% level; * significant at 10% level.

Table 8

Ex	port Decisi	on: Probit E	stimates w	ith Correct	ion for Pote	ntial Samp	le Selection	1 Bias		
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)
Export(-1)	2.613***	1.960***	1.936***	1.917***	1.936***	1.936***	2.120***	2.101***	2.117***	2.075***
	(0.208)	(0.215)	(0.187)	(0.190)	(0.194)	(0.193)	(0.237)	(0.239)	(0.239)	(0.243)
Employment		0.512***	0.528***	0.511***	0.516***	0.538***	0.710***	0.723***	0.713***	0.824***
		(0.131)	(0.130)	(0.135)	(0.140)	(0.146)	(0.172)	(0.173)	(0.174)	(0.176)
Age			-0.092	-0.079	-0.059	-0.056	-0.044	-0.031	-0.030	-0.029
			(0.131)	(0.123)	(0.122)	(0.124)	(0.122)	(0.125)	(0.126)	(0.130)
Training				0.114	0.149	0.156	0.181	0.196	0.199	0.094
				(0.179)	(0.179)	(0.180)	(0.219)	(0.227)	(0.228)	(0.234)
Domestic Inputs					-0.597**	-0.582*	-0.817**	-0.813**	-0.861***	-0.714**
					(0.301)	(0.301)	(0.325)	(0.323)	(0.331)	(0.331)
Group						-0.163	-0.073	-0.097	-0.033	-0.180
						(0.246)	(0.268)	(0.271)	(0.296)	(0.317)
Product Improvement							0.459***	0.462***	0.472***	0.551***
							(0.177)	(0.177)	(0.175)	(0.171)
Product Introduction								-0.177	-0.164	-0.126
T N. 11								(0.227)	(0.228)	(0.231)
Exporting Neighbors									0.273	0.335
Calaa maa Emanlaria									(0.301)	(0.295) 0.448***
Sales per Employee										(0.148)
	1	2/	27	2/	2/	2/	27	27		
Industry Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Region Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Year Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Observations	376	367	367	367	367	367	330	330	330	328

The table reports probit estimates of Equation (7) for the period 1997-1998 controlling for potential sample selection bias as indicated in the main text (Page 10). Export is a binary variable taking the value of 1 if the firm exports and 0 otherwise (Export(-1) is the one-year lagged value). Employment is the natural logarithm of the firm employment level (lagged one year). Age is the natural logarithm of the firm's age (lagged one year). Training is a binary variable taking the value of 1 if the firm has performed training activities for its employees the previous year and 0 otherwise. Domestic Inputs is a binary variable taking the value of 1 if the firm sources its inputs mainly from domestic providers, i.e., over 50% and 0 otherwise (lagged one year). Group is a binary variable taking the value of 1 if the firm belongs to a larger economic group and 0 otherwise. Product Improvement is a binary variable taking the value of 1 if the firm has invested in changing and improving the quality of its already existent product the previous year and 0 otherwise. Product Introduction is a binary variable taking the value of 1 if the firm has invested in introducing new products into the market in the previous year and 0 otherwise. Exporting Neighbors is a binary variable taking the value of 1 if there has been at least one exporting firm in the same industry in the same province the previous year and 0 otherwise. Sales per Employee is the natural logarithm of the firm's total sales divided by the number of employees (lagged one year). Standard errors accounting for stratification by region are reported below estimated coefficients between parentheses. *** significant at 1% level; ** significant at 5% level; * significant at 10% level.

Table 9

	Expo	rt Decision	: Probit Est	imates witl	ı Bootstrap	ped Standa	rd Errors			
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)
Export(-1)	2.132***	1.942***	1.936***	1.917***	1.936***	1.936***	2.118***	2.105***	2.118***	2.059***
	(0.206)	(0.220)	(0.238)	(0.245)	(0.253)	(0.245)	(0.308)	(0.319)	(0.312)	(0.268)
Employment		0.511***	0.528***	0.511***	0.516***	0.538***	0.716***	0.718***	0.715***	0.828***
		(0.151)	(0.151)	(0.169)	(0.167)	(0.182)	(0.196)	(0.218)	(0.216)	(0.236)
Age			-0.089	-0.079	-0.059	-0.056	-0.073	-0.065	-0.064	-0.074
			(0.143)	(0.147)	(0.151)	(0.151)	(0.162)	(0.172)	(0.173)	(0.179)
Training				0.114	0.149	0.156	0.138	0.152	0.155	0.036
				(0.213)	(0.210)	(0.212)	(0.248)	(0.264)	(0.257)	(0.267)
Domestic Inputs					-0.597*	-0.582	-0.895***	-0.890**	-0.934***	-0.801**
					(0.361)	(0.376)	(0.403)	(0.434)	(0.396)	(0.429)
Group						-0.163	-0.131	-0.146	-0.096	-0.259
						(0.306)	(0.330)	(0.369)	(0.357)	(0.383)
Product Improvement							0.450**	0.452*	0.456*	0.532***
							(0.241)	(0.263)	(0.258)	(0.254)
Product Introduction								-0.118	-0.109	-0.072
								(0.275)	(0.285)	(0.304)
Exporting Neighbors									0.266	0.362
Calanara Faranta									(0.454)	(0.491)
Sales per Employee										0.471*** (0.218)
					.,					
Industry Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Region Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Year Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Pseudo-R ²	0.400	0.421	0.422	0.423	0.433	0.434	0.477	0.477	0.478	0.497
Observations	376	367	367	367	367	367	330	330	330	328

The table reports probit estimates of Equation (7) for the period 1997-1998. Export is a binary variable taking the value of 1 if the firm exports and 0 otherwise (Export(-1) is the one-year lagged value). Employment is the natural logarithm of the firm employment level (lagged one year). Age is the natural logarithm of the firm's age (lagged one year). Training is a binary variable taking the value of 1 if the firm has performed training activities for its employees the previous year and 0 otherwise. Domestic Inputs is a binary variable taking the value of 1 if the firm sources its inputs mainly from domestic providers, i.e., over 50% and 0 otherwise (lagged one year). Group is a binary variable taking the value of 1 if the firm has invested in changing and improving the quality of its already existent product the previous year and 0 otherwise. Product Introduction is a binary variable taking the value of 1 if the firm has invested in introducing new products into the market in the previous year and 0 otherwise. Exporting Neighbors is a binary variable taking the value of 1 if there has been at least one exporting firm in the same industry in the same province the previous year and 0 otherwise. Sales per Employee is the natural logarithm of the firm's total sales divided by the number of employees (lagged one year). Bootstrapped standard errors are reported below estimated coefficients between parentheses (based on 1,000 replications). **** significant at 1% level; ** significant at 5% level; * significant at 10% level.

Table 10

	Exp	ort Decisio	n: Probit E	stimates for	Non-Swite	chers			
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)
Employment	0.982***	0.999***	0.876***	0.875***	0.879***	0.968***	0.993***	0.993***	1.224***
	(0.233)	(0.238)	(0.245)	(0.244)	(0.251)	(0.289)	(0.301)	(0.301)	(0.318)
Age		-0.114	-0.060	-0.053	-0.052	0.034	0.041	0.043	0.040
		(0.172)	(0.174)	(0.174)	(0.173)	(0.180)	(0.184)	(0.185)	(0.198)
Training			0.635***	0.628***	0.629***	0.455*	0.483**	0.481**	0.295
			(0.233)	(0.234)	(0.234)	(0.236)	(0.239)	(0.239)	(0.250)
Domestic Inputs				-0.190	-0.185	-0.445*	-0.422	-0.446*	-0.244
				(0.252)	(0.252)	(0.258)	(0.263)	(0.263)	(0.288)
Group					-0.043	0.098	-0.026	-0.007	-0.080
					(0.316)	(0.304)	(0.320)	(0.321)	(0.351)
Product Improvement						0.221	0.293*	0.294*	0.387**
						(0.181)	(0.172)	(0.172)	(0.177)
Product Introduction							-0.470*	-0.465*	-0.412
							(0.253)	(0.256)	(0.267)
Exporting Neighbors								0.120	0.167
								(0.431)	(0.507)
Sales per Employee									0.752***
									(0.209)
Industry Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Region Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Year Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Pseudo-R ²	0.277	0.278	0.305	0.306	0.306	0.313	0.327	0.328	0.328
Observations	258	258	258	258	258	231	231	231	. 229

The table reports probit estimates of Equation (7) for the period 1997-1998 based on a sample that only includes non-switching firms. Export is a binary variable taking the value of 1 if the firm exports and 0 otherwise (Export(-1) is the one-year lagged value). Employment is the natural logarithm of the firm employment level (lagged one year). Age is the natural logarithm of the firm's age (lagged one year). Training is a binary variable taking the value of 1 if the firm has performed training activities for its employees the previous year and 0 otherwise. Domestic Inputs is a binary variable taking the value of 1 if the firm sources its inputs mainly from domestic providers, i.e., over 50% and 0 otherwise (lagged one year). Group is a binary variable taking the value of 1 if the firm belongs to a larger economic group and 0 otherwise. Product Improvement is a binary variable taking the value of 1 if the firm has invested in changing and improving the quality of its already existent product the previous year and 0 otherwise. Product Introduction is a binary variable taking the value of 1 if the firm has invested in introducing new products into the market in the previous year and 0 otherwise. Exporting Neighbors is a binary variable taking the value of 1 if there has been at least one exporting firm in the same industry in the same province the previous year and 0 otherwise. Sales per Employee is the natural logarithm of the firm's total sales divided by the number of employees (lagged one year). Robust standard errors clustered on firm are reported below estimated coefficients between parentheses. *** significant at 1% level; ** significant at 5% level; * significant at 10% level.

Table 11

	(1)	(2)	Export Decision: Random-Effects Probit Estimates										
		(2)	(3)	(4)									
Export(-1)	2.118***	2.105***	2.118***	2.059***									
- ' '	(0.220)	(0.221)	(0.224)	(0.229)									
Employment	0.716***	0.718***	0.715***	0.828***									
	(0.164)	(0.164)	(0.165)	(0.177)									
Age	-0.073	-0.065	-0.064	-0.074									
	(0.130)	(0.130)	(0.130)	(0.133)									
Training	0.138	0.152	0.155	0.036									
	(0.205)	(0.207)	(0.207)	(0.217)									
Domestic Inputs	-0.895***	-0.890***	-0.934***	-0.801**									
	(0.303)	(0.304)	(0.314)	(0.217)									
Group	-0.131	-0.146	-0.096	-0.260									
	(0.291)	(0.292)	(0.302)	(0.319)									
Product Improvement	0.450**	0.452**	0.456***	0.532**									
	(0.215)	(0.215)	(0.215)	(0.221)									
Product Introduction		-0.118	-0.109	-0.072									
		(0.204)	(0.205)	(0.210)									
Exporting Neighbors			0.266	0.362									
			(0.385)	(0.400)									
Sales per Employee				0.471***									
				(0.175)									
Industry Fixed Effects	Yes	Yes	Yes	Yes									
Year Fixed Effects	Yes	Yes	Yes	Yes									
Region Fixed Effects	Yes	Yes	Yes	Yes									
Likelihood Ratio Test	0.000	0.000	0.000	0.000									
	[1.000]	[1.000]	[1.000]	[1.000]									
Observations	338	338	338	336									

The table reports random-effects probit estimates of Equation (8) for the period 1997-1998. Export is a binary variable taking the value of 1 if the firm exports and 0 otherwise (Export(-1) is the one-year lagged value). Employment is the natural logarithm of the firm employment level (lagged one year). Age is the natural logarithm of the firm's age (lagged one year). Training is a binary variable taking the value of 1 if the firm has performed training activities for its employees the previous year and 0 otherwise. Domestic Inputs is a binary variable taking the value of 1 if the firm sources its inputs mainly from domestic providers, i.e., over 50% and 0 otherwise (lagged one year). Group is a binary variable taking the value of 1 if the firm belongs to a larger economic group and 0 otherwise. Product Improvement is a binary variable taking the value of 1 if the firm has invested in changing and improving the quality of its already existent product the previous year and 0 otherwise. Product Introduction is a binary variable taking the value of 1 if the firm has invested in introducing new products into the market in the previous year and 0 otherwise. Exporting Neighbors is a binary variable taking the value of 1 if there has been at least one exporting firm in the same industry in the same province the previous year and 0 otherwise. Sales per Employee is the natural logarithm of the firm's total sales divided by the number of employees (lagged one year). Standard errors are reported below estimated coefficients between parentheses. *** significant at 1% level; ** significant at 5% level; * significant at 10% level.

Table 12

Ех	cport Decision by I	Jestination	Markets: Seen	ningly Unre	lated Bivariat	e Probit		
	(1)		(2)		(3)		(4)
	MERC	ROW	MERC	ROW	MERC	ROW	MERC	ROW
Export(-1)	1.920***	1.815***	1.904***	1.817***	1.923***	1.818***	1.917***	1.804***
	(0.217)	(0.225)	(0.220)	(0.227)	(0.222)	(0.227)	(0.230)	(0.229)
Employment	0.744***	0.541***	0.748***	0.541***	0.738***	0.539***	0.887***	0.551***
	(0.170)	(0.172)	(0.170)	(0.172)	(0.171)	(0.172)	(0.190)	(0.175)
Age	-0.170	-0.020	-0.165	-0.021	-0.166	-0.020	-0.195	-0.017
_	(0.135)	(0.147)	(0.135)	(0.147)	(0.136)	(0.147)	(0.140)	(0.147)
Training	0.074	0.575**	0.083	0.573**	0.080	0.573**	-0.088	0.559**
	(0.212)	(0.226)	(0.213)	(0.227)	(0.214)	(0.227)	(0.229)	(0.228)
Domestic Inputs	-0.754***	-0.409	-0.750***	-0.408	-0.789***	-0.415	-0.608**	-0.386
	(0.288)	(0.305)	(0.289)	(0.305)	(0.297)	(0.312)	(0.299)	(0.315)
Group	-0.350	0.197	-0.365	0.197	-0.317	0.207	-0.547*	0.177
	(0.288)	(0.276)	(0.291)	(0.277)	(0.298)	(0.283)	(0.319)	(0.290)
Product Improvement	0.324	0.064	0.327	0.066	0.330	0.067	0.397*	0.082
	(0.218)	(0.222)	(0.218)	(0.222)	(0.218)	(0.222)	(0.223)	(0.222)
Product Introduction			-0.081	0.014	-0.067	0.017	-0.037	0.019
			(0.204)	(0.225)	(0.205)	(0.227)	(0.210)	(0.229)
Exporting Neighbors					0.298	0.064	0.364	0.069
					(0.407)	(0.398)	(0.416)	(0.402)
Sales per Employee							0.498***	0.093
							(0.181)	(0.147)
Industry Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Region Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Year Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Wald Test Rho=0	8.997	***	8.802	***	8.803	***	9.578	3***
	[0.00	2]	[0.00	3]	[0.00	3]	[0.0]	02]
Observations	338	338	338	338	338	338	336	336

The table reports seemingly unrelated bivariate probit estimates of Equations (9)-(10) for the period 1997-1998. Export is a binary variable taking the value of 1 if the firm exports and 0 otherwise (Export(-1) is the one-year lagged value). Employment is the natural logarithm of the firm employment level (lagged one year). Age is the natural logarithm of the firm's age (lagged one year). Training is a binary variable taking the value of 1 if the firm has performed training activities for its employees the previous year and 0 otherwise. Domestic Inputs is a binary variable taking the value of 1 if the firm sources its inputs mainly from domestic providers, i.e., over 50% and 0 otherwise (lagged one year). Group is a binary variable taking the value of 1 if the firm belongs to a larger economic group and 0 otherwise. Product Improvement is a binary variable taking the value of 1 if the firm has invested in changing and improving the quality of its already existent product the previous year and 0 otherwise. Product Introduction is a binary variable taking the value of 1 if the firm has invested in introducing new products into the market in the previous year and 0 otherwise. Exporting Neighbors is a binary variable taking the value of 1 if there has been at least one exporting firm in the same industry in the same province the previous year and 0 otherwise. Sales per Employee is the natural logarithm of the firm's total sales divided by the number of employees (lagged one year). Standard errors are reported below estimated coefficients between parentheses. *** significant at 1% level; ** significant at 5% level; * significant at 10% level.

Appendix

Table A1

Pairwise Correlation between Main Explanatory Variables										
Variables	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)
(1) Export(-1)	1.000									
(2) Employment	0.303	1.000								
(3) Age	0.000	0.104	1.000							
(4) Training	0.243	0.247	-0.065	1.000						
(5) Domestic Inputs	-0.102	0.029	0.084	0.105	1.000					
(6) Group	0.136	0.320	0.043	0.096	0.016	1.000				
(7) Product Improvement	0.027	-0.010	-0.026	0.068	0.144	0.012	1.000			
(8) Product Introduction	-0.039	0.031	0.030	0.134	0.176	-0.097	0.056	1.000		
(9) Exporting Neighbors	0.118	-0.023	0.069	-0.023	0.035	-0.084	0.021	-0.030	1.000	
(10) Sales per Employee	0.175	-0.025	-0.052	0.158	-0.118	0.120	-0.073	0.005	-0.028	1.000

Export is a binary variable taking the value of 1 if the firm exports and 0 otherwise (Export(-1) is the one-year lagged value). Employment is the natural logarithm of the firm employment level (lagged one year). Age is the natural logarithm of the firm's age (lagged one year). Training is a binary variable taking the value of 1 if the firm has performed training activities for its employees the previous year and 0 otherwise. Domestic Inputs is a binary variable taking the value of 1 if the firm sources its inputs mainly from domestic providers, i.e., over 50% and 0 otherwise (lagged one year). Group is a binary variable taking the value of 1 if the firm belongs to a larger economic group and 0 otherwise. Product Improvement is a binary variable taking the value of 1 if the firm has invested in changing and improving the quality of its already existent product the previous year and 0 otherwise. Product Introduction is a binary variable taking the value of 1 if the firm has invested in introducing new products into the market in the previous year and 0 otherwise. Exporting Neighbors is a binary variable taking the value of 1 if there has been at least one exporting firm in the same industry in the same province the previous year and 0 otherwise. Sales per Employee is the natural logarithm of the firm's total sales divided by the number of employees (lagged one year).

Table A2

		Export	Decision: l	Probit Estir	nates - Mar	ginal Effec	ts			
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)
Export(-1)	0.704***	0.657***	0.655***	0.650***	0.656***	0.656***	0.699***	0.696***	0.699***	0.685***
	(0.046)	(0.052)	(0.052)	(0.053)	(0.054)	(0.054)	(0.052)	(0.053)	(0.053)	(0.055)
Employment		0.199***	0.205***	0.198***	0.201***	0.209***	0.279***	0.281***	0.279***	0.323***
		(0.048)	(0.050)	(0.051)	(0.053)	(0.054)	(0.065)	(0.066)	(0.066)	(0.065)
Age			-0.035	-0.031	-0.023	-0.021	-0.028	-0.025	-0.025	-0.029
			(0.041)	(0.042)	(0.042)	(0.042)	(0.047)	(0.048)	(0.048)	(0.051)
Training				0.044	0.058	0.060	0.054	0.059	0.060	0.014
				(0.071)	(0.072)	(0.073)	(0.081)	(0.083)	(0.083)	(0.087)
Domestic Inputs					-0.230**	-0.224**	-0.340***	-0.338***	-0.354***	-0.306***
					(0.113)	(0.113)	(0.112)	(0.112)	(0.111)	(0.115)
Group						-0.062	-0.050	-0.056	-0.037	-0.099
						(0.083)	(0.090)	(0.091)	(0.097)	(0.101)
Product Improvement							0.177***	0.177***	0.179***	0.209***
							(0.069)	(0.069)	(0.068)	(0.065)
Product Introduction								-0.046	-0.042	-0.028
T N. 11								(0.084)	(0.085)	(0.085)
Exporting Neighbors									0.102	0.138
Calaa man Emanlaria									(0.111)	(0.108) 0.184***
Sales per Employee										(0.057)
	27	2/	1/	1/		37		37		
Industry Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Region Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Year Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Pseudo-R ²	0.400	0.421	0.422	0.423	0.433	0.434	0.477	0.477	0.478	0.497
Observations	376	367	367	367	367	367	330	330	330	328

The table reports marginal effects obtained from probit estimates of Equation (7) for the period 1997-1998. Export is a binary variable taking the value of 1 if the firm exports and 0 otherwise (Export(-1) is the one-year lagged value). Employment is the natural logarithm of the firm employment level (lagged one year). Age is the natural logarithm of the firm's age (lagged one year). Training is a binary variable taking the value of 1 if the firm has performed training activities for its employees the previous year and 0 otherwise. Domestic Inputs is a binary variable taking the value of 1 if the firm sources its inputs mainly from domestic providers, i.e., over 50% and 0 otherwise (lagged one year). Group is a binary variable taking the value of 1 if the firm has invested in changing and improving the quality of its already existent product the previous year and 0 otherwise. Product Introduction is a binary variable taking the value of 1 if the firm has invested in introducing new products into the market in the previous year and 0 otherwise. Exporting Neighbors is a binary variable taking the value of 1 if there has been at least one exporting firm in the same industry in the same province the previous year and 0 otherwise. Sales per Employee is the natural logarithm of the firm's total sales divided by the number of employees (lagged one year). Robust standard errors clustered on firm are reported below estimated coefficients between parentheses. *** significant at 1% level; ** significant at 5% level; * significant at 10% level.

NOTE DI LAVORO DELLA FONDAZIONE ENI ENRICO MATTEI

Fondazione Eni Enrico Mattei Working Paper Series

Our Note di Lavoro are available on the Internet at the following addresses:

http://www.feem.it/Feem/Pub/Publications/WPapers/default.htm http://www.ssrn.com/link/feem.html http://www.repec.org http://agecon.lib.umn.edu http://www.bepress.com/feem/

	4 0000	NOTE DI LAVORO PUBLISHED IN 2009
SD	1.2009	Michael Hoel: <u>Bush Meets Hotelling: Effects of Improved Renewable Energy Technology on Greenhouse Gas</u>
CD	2 2000	Emissions
SD	2.2009	Abay Mulatu, Reyer Gerlagh, Dan Rigby and Ada Wossink: Environmental Regulation and Industry Location
SD	3.2009	Anna Alberini, Stefania Tonin and Margherita Turvani: Rates of Time Preferences for Saving Lives in the Hazardous Waste Site Context
SD	4.2009	Elena Ojea, Paulo A.L.D. Nunes and Maria Loureiro: Mapping of Forest Biodiversity Values: A Plural
30	4.2007	Perspective
SD	5.2009	Xavier Pautrel: Macroeconomic Implications of Demography for the Environment: A Life-Cycle Perspective
IM	6.2009	Andrew Ellul, Marco Pagano and Fausto Panunzi: Inheritance Law and Investment in Family Firms
IM	7.2009	Luigi Zingales: The Future of Securities Regulation
SD	8.2009	Carlo Carraro, Emanuele Massetti and Lea Nicita: How Does Climate Policy Affect Technical Change? An
		Analysis of the Direction and Pace of Technical Progress in a Climate-Economy Model
SD	9.2009	William K. Jaeger: The Welfare Effects of Environmental Taxation
SD	10.2009	Aude Pommeret and Fabien Prieur: Double Irreversibility and Environmental Policy Design
SD	11.2009	Massimiliano Mazzanti and Anna Montini: Regional and Sector Environmental Efficiency Empirical Evidence
		from Structural Shift-share Analysis of NAMEA data
SD	12.2009	A. Chiabai, C. M. Travisi, H. Ding, A. Markandya and P.A.L.D Nunes: <u>Economic Valuation of Forest</u>
		Ecosystem Services: Methodology and Monetary Estimates
SD	13.2009	Andrea Bigano, Mariaester Cassinelli, Fabio Sferra, Lisa Guarrera, Sohbet Karbuz, Manfred Hafner, Anil
CD	4.4.0000	Markandya and Ståle Navrud: The External Cost of European Crude Oil Imports
SD	14.2009	Valentina Bosetti, Carlo Carraro, Romain Duval, Alessandra Sgobbi and Massimo Tavoni: The Role of R&D
18.4	15 2000	and Technology Diffusion in Climate Change Mitigation: New Perspectives Using the Witch Model
IM	15.2009	Andrea Beltratti, Marianna Caccavaio and Bernardo Bortolotti: Stock Prices in a Speculative Market: The
GC	16.2009	Chinese Split-Share Reform Angelo Antoci, Fabio Sabatini and Mauro Sodini: The Fragility of Social Capital
SD	17.2009	Alexander Golub, Sabine Fuss, Jana Szolgayova and Michael Obersteiner: Effects of Low-cost Offsets on
30	17.2009	Energy Investment – New Perspectives on REDD –
SD	18.2009	Enrica De Cian: Factor-Augmenting Technical Change: An Empirical Assessment
SD	19.2009	Irene Valsecchi: Non-Uniqueness of Equilibria in One-Shot Games of Strategic Communication
SD	20.2009	Dimitra Vouvaki and Anastasios Xeapapadeas: Total Factor Productivity Growth when Factors of Production
30	20.2003	Generate Environmental Externalities
SD	21.2009	Giulia Macagno, Maria Loureiro, Paulo A.L.D. Nunes and Richard Tol: Assessing the Impact of Biodiversity
		on Tourism Flows: A model for Tourist Behaviour and its Policy Implications
IM	22.2009	Bernardo Bortolotti, Veljko Fotak, William Megginson and William Miracky: Sovereign Wealth Fund
		Investment Patterns and Performance
IM	23.2009	Cesare Dosi and Michele Moretto: Auctioning Monopoly Franchises: Award Criteria and Service Launch
		<u>Requirements</u>
SD	24.2009	Andrea Bastianin: Modelling Asymmetric Dependence Using Copula Functions: An application to Value-at-
		Risk in the Energy Sector
IM	25.2009	Shai Bernstein, Josh Lerner and Antoinette Schoar: The Investment Strategies of Sovereign Wealth Funds
SD	26.2009	Marc Germain, Henry Tulkens and Alphonse Magnus: <u>Dynamic Core-Theoretic Cooperation in a Two-</u>
18.4	27 2000	Dimensional International Environmental Model French Partners Overdeness design of Cardia Parties Was a Primary Course of the Criticis
IM SD	27.2009	Frank Partnoy: <u>Overdependence on Credit Ratings Was a Primary Cause of the Crisis</u> Frank H. Page Jr and Myrna H. Wooders (lxxxv): <u>Endogenous Network Dynamics</u>
SD	28.2009 29.2009	Caterina Calsamiglia, Guillaume Haeringer and Flip Klijnb (lxxxv): Constrained School Choice: An
30	29.2009	Experimental Study
SD	30.2009	Gilles Grandjean, Ana Mauleon and Vincent Vannetelbosch (lxxxv): Connections Among Farsighted Agents
SD	31.2009	Antonio Nicoló and Carmelo Rodríguez Álvarez (lxxxv): Feasibility Constraints and Protective Behavior in
55	01.2007	Efficient Kidney Exchange
SD	32.2009	Rahmi İlkiliç (İxxxv): <u>Cournot Competition on a Network of Markets and Firms</u>
SD	33.2009	Luca Dall'Asta, Paolo Pin and Abolfazl Ramezanpour (lxxxv): Optimal Equilibria of the Best Shot Game
SD	34.2009	Edoardo Gallo (lxxxv): Small World Networks with Segregation Patterns and Brokers
SD	35.2009	Benjamin Golub and Matthew O. Jackson (lxxxv): How Homophily Affects Learning and Diffusion in
		Networks

CD	26 2000	Mades Mark de Alexander Transfer Services
SD SD	36.2009 37.2009	Markus Kinateder (lxxxv): <u>Team Formation in a Network</u> Constanza Fosco and Friederike Mengel (lxxxv): <u>Cooperation through Imitation and Exclusion in Networks</u>
	38.2009	
SD		Berno Buechel and Tim Hellmann (lxxxv): <u>Under-connected and Over-connected Networks</u>
SD	39.2009	Alexey Kushnir (lxxxv): Matching Markets with Signals
SD	40.2009	Alessandro Tavoni (lxxxv): Incorporating Fairness Motives into the Impulse Balance Equilibrium and Quantal Response Equilibrium Concepts: An Application to 2x2 Games
SD	41.2009	Steven J. Brams and D. Marc Kilgour (lxxxv): <u>Kingmakers and Leaders in Coalition Formation</u>
SD		
30	42.2009	Dotan Persitz (lxxxv): <u>Power in the Heterogeneous Connections Model: The Emergence of Core-Periphery</u> Networks
SD	43.2009	Fabio Eboli, Ramiro Parrado, Roberto Roson: <u>Climate Change Feedback on Economic Growth: Explorations</u>
30	43.2009	with a Dynamic General Equilibrium Mode
GC	44.2009	Fabio Sabatini: Does Social Capital Create Trust? Evidence from a Community of Entrepreneurs
SD	45.2009	ZhongXiang Zhang: Is it Fair to Treat China as a Christmas Tree to Hang Everybody's Complaints? Putting
35	10.2005	its Own Energy Saving into Perspective
SD	46.2009	Eftichios S. Sartzetakis, Anastasios Xepapadeas and Emmanuel Petrakis: The Role of Information Provision
		as a Policy Instrument to Supplement Environmental Taxes: Empowering Consumers to Choose Optimally
SD	47.2009	Jean-François Caulier, Ana Mauleon and Vincent Vannetelbosch: Contractually Stable Networks
GC	48.2009	Massimiliano Mazzanti, Susanna Mancinelli, Giovanni Ponti and Nora Piva: Education, Reputation or
		Network? Evidence from Italy on Migrant Workers Employability
SD	49.2009	William Brock and Anastasios Xepapadeas: General Pattern Formation in Recursive Dynamical Systems
		Models in Economics
SD	50.2009	Giovanni Marin and Massimiliano Mazzanti: Emissions Trends and Labour Productivity Dynamics Sector
		Analyses of De-coupling/Recoupling on a 1990-2005 Namea
SD	51.2009	Yoshio Kamijo and Ryo Kawasaki (lxxxv): <u>Dynamics, Stability, and Foresight in the Shapley-Scarf Housing</u>
		<u>Market</u>
IM	52.2009	Laura Poddi and Sergio Vergalli: Does Corporate Social Responsibility Affect the Performance of Firms?
SD	53.2009	Valentina Bosetti, Carlo Carraro and Massimo Tavoni: Climate Change Mitigation Strategies in Fast-
66	54.0000	Growing Countries: The Benefits of Early Action
GC	54.2009	Alireza Naghavi and Gianmarco I.P. Ottaviano: Firm Heterogeneity, Contract Enforcement, and the Industry
15.4	<i>EE</i> 2000	Dynamics of Offshoring Ciarran Calladaria and Carla Sarran Carladaria and Carrantition Break and Carra & Discretified
IM	55.2009	Giacomo Calzolari and Carlo Scarpa: On Regulation and Competition: Pros and Cons of a Diversified Monopolist
SD	56.2009	Valentina Bosetti, Ruben Lubowski and Alexander Golub and Anil Markandya: <u>Linking Reduced</u>
30	30.2009	Deforestation and a Global Carbon Market: Impacts on Costs, Financial Flows, and Technological
		Innovation
IM	57.2009	Emmanuel Farhi and Jean Tirole: Collective Moral Hazard, Maturity Mismatch and Systemic Bailouts
SD	58.2009	Kelly C. de Bruin and Rob B. Dellink: How Harmful are Adaptation Restrictions
SD	59.2009	Rob Dellink, Michel den Elzen, Harry Aiking, Emmy Bergsma, Frans Berkhout, Thijs Dekker, Joyeeta Gupta:
SD	37.2007	Sharing the Burden of Adaptation Financing: An Assessment of the Contributions of Countries
SD	60.2009	Stefania Tonin, Anna Alberini and Margherita Turvani: The Value of Reducing Cancer Risks at Contaminated
		Sites: Are More Heavily Exposed People Willing to Pay More?
SD	61.2009	Clara Costa Duarte, Maria A. Cunha-e-Sá and Renato Rosa: The Role of Forests as Carbon Sinks: Land-Use
		and Carbon Accounting
GC	62.2009	Carlo Altomonte and Gabor Békés: <u>Trade Complexity and Productivity</u>
GC	63.2009	Elena Bellini, Gianmarco I.P. Ottaviano, Dino Pinelli and Giovanni Prarolo: Cultural Diversity and Economic
		Performance: Evidence from European Regions
SD	64.2009	Valentina Bosetti, Carlo Carraro, Enrica De Cian, Romain Duval, Emanuele Massetti and Massimo Tavoni:
		The Incentives to Participate in, and the Stability of, International Climate Coalitions: A Game-theoretic
		Analysis Using the Witch Model
IM	65.2009	John Temple Lang: <u>Article 82 EC - The Problems and The Solution</u>
SD	66.2009	P. Dumas and S. Hallegatte: Think Again: Higher Elasticity of Substitution Increases Economic Resilience
SD	67.2009	Ruslana Rachel Palatnik and Roberto Roson: Climate Change Assessment and Agriculture in General
		Equilibrium Models: Alternative Modeling Strategies
SD	68.2009	Paulo A.L.D. Nunes, Helen Ding and Anil Markandya: The Economic Valuation of Marine Ecosystems
IM	69.2009	Andreas Madestam: Informal Finance: A Theory of Moneylenders
SD	70.2009	Efthymia Kyriakopoulou and Anastasios Xepapadeas: Environmental Policy, Spatial Spillovers and the
		Emergence of Economic Agglomerations
SD	71.2009	A. Markandya, S. Arnold, M. Cassinelli and T. Taylor: <u>Coastal Zone Management in the Mediterranean:</u>
		Legal and Economic Perspectives
GC	72.2009	Gianmarco I.P. Ottaviano and Giovanni Prarolo: <u>Cultural Identity and Knowledge Creation in Cosmopolitan</u>
-		<u>Cities</u>
SD	73.2009	Erik Ansink: Self-enforcing Agreements on Water allocation
GC	74.2009	Mario A. Maggioni, Francesca Gambarotto and T. Erika Uberti: Mapping the Evolution of "Clusters": A
-		Meta-analysis
SD	75.2009	Nektarios Aslanidis: Environmental Kuznets Curves for Carbon Emissions: A Critical Survey
SD	76.2009	Joan Canton: Environmentalists' Behaviour and Environmental Policies
SD		
30	77.2009	Christoph M. Rheinberger: Paying for Safety: Preferences for Mortality Risk Reductions on Alpine Roads
30	77.2009	Christoph M. Rheinberger: Paying for Safety: Preferences for Mortality Risk Reductions on Alpine Roads

IM	78.2009	Chiara D'Alpaos, Michele Moretto, Paola Valbonesi and Sergio Vergalli: "It Is Never too late": Optimal
		Penalty for Investment Delay in Public Procurement Contracts
SD	79.2009	Henry Tulkens and Vincent van Steenberghe: "Mitigation, Adaptation, Suffering": In Search of the Right Mix
		in the Face of Climate Change
SD	80.2009	Giovanni Bella: <u>A Search Model for Joint Implementation</u>
SD	81.2009	ZhongXiang Zhang: Multilateral Trade Measures in a Post-2012 Climate Change Regime?: What Can Be
		Taken from the Montreal Protocol and the WTO?
SD	82.2009	Antoine Dechezleprêtre, Matthieu Glachant, Ivan Hascic, Nick Johnstone and Yann Ménière: Invention and
		Transfer of Climate Change Mitigation Technologies on a Global Scale: A Study Drawing on Patent Data
SD	83.2009	László Á. Kóczy: Stationary Consistent Equilibrium Coalition Structures Constitute the Recursive Core
SD	84.2009	Luca Di Corato and Michele Moretto: Investing in Biogas: Timing, Technological Choice and the Value of
		Flexibility from Inputs Mix
SD	85.2009	Valentina Bosetti, Enrica De Cian, Alessandra Sgobbi, and Massimo Tavoni: The 2008 WITCH Model: New
		Model Features and Baseline
IM	86.2009	Rocco Macchiavello: Vertical Integration and Investor Protection in Developing Countries
SD	87.2009	Massimiliano Mazzanti and Antonio Musolesi: Carbon Kuznets Curves: Long-run Structural Dynamics and
		Policy Events
GC	88.2009	Gianmarco I.P. Ottaviano and Christian Volpe Martincus: SMEs in Argentina: Who are the Exporters

(lxxxv) This paper has been presented at the 14th Coalition Theory Network Workshop held in Maastricht, The Netherlands, on 23-24 January 2009 and organised by the Maastricht University CTN group (Department of Economics, http://www.feem-web.it/ctn/12d_maa.php).